Author's response to reviews

Title: Post-viral polyneuropathy with parasympathetic dysautonomia in a child: a case report

Authors:

Terence F McLoughlin (mda05tfm@shef.ac.uk)
Adelaida Martinez (adelaida_martinez@hotmail.com)

Version: 8 Date: 28 March 2010

Author's response to reviews: see over
Dear Editor,

Thank you for editing my case report entitled Post-adenoviral parasympathetic dysautonomia in a child: a case report.

I have made the following improvements that you suggested:

- The running title includes the word “report” instead of “presentation”
- Corrected author details
- Ethnic background of patient included
- Common causative agents and more details on investigation included
- An abstract comprising of 3 sections: Intro, case presentation and conclusion
- The quality of the language used in the report has improved

Yours sincerely,

Terence McLoughlin